Author's response to reviews

Title: Rationale and Methods of the Multicenter Randomised Trial of a Heart Failure Management Programme among Geriatric Patients (HF-Geriatrics)

Authors:

Carlos Rodriguez-Pascual (carlos.rodriguez.pascual@sergas.es)
Emilio Paredes-Galan (eparedesgalan@hotmail.com)
Jose Luis Gonzalez-Guerrero (jglezg@teleline.es)
Rocio Menendez-Colino (rociocolino@hotmail.com)
Pedro Abizanda-Soler (pabizanda@sescam.jccm.es)
Mercedes Horillos-Calvo (mhornillos@sescam.jccm.es)
Juan Jose Solano Jaurieta (juanjose.solano@sespa.princast.es)
jorge Manzarbeitia-Arambarri (jmanzarbeitia.hugf@salud.madrid.org)
Jose Manuel Ribera-Casado (jribera.hcsc@salud.madrid.org)
Fernando Rodriguez-Artalejo (fernando.artalejo@uam.es)

Version: 2 Date: 26 July 2011

Author's response to reviews: see over
Dear Editor,

Please, find enclosed a new version of the manuscript entitled “Rationale and Methods of the Multicenter Randomised Trial of a Heart Failure Management Programme among Geriatric Patients (HF-Geriatrics)”, to be considered for publication as an study protocol in BMC Public Health.

The following changes has been made to the manuscript

(1) A title page with the author’s full names, institutional addresses, and e-mail have been added. Details related with the corresponding author are also indicated.

(2) The specific names of the ethics committees which approved the study have been described in a new section at the beginning of the manuscript, following the title page.

We thank you for your attention in this matter.

Yours faithfully,

Carlos Rodríguez Pascual MD, PhD

Geriatric Medicine Department
Hospital Meixoeiro
Complejo Hospitalario Universitario de Vigo
36200 Vigo
Spain